Author’s response to reviews

Title: A randomised controlled trial testing the effects of cognitive behavioural therapy for insomnia on the mental health of university students

Authors:

Daniel Freeman (daniel.freeman@psych.ox.ac.uk)
Bryony Sheaves (bryony.sheaves@psych.ox.ac.uk)
Guy Goodwin (guy.goodwin@psych.ox.ac.uk)
Ly-Mee Yu (ly-mee.yu@phc.ox.ac.uk)
Paul Harrison (Paul.harrison@psych.ox.ac.uk)
Richard Emsley (Richard.Emsley@manchester.ac.uk)
Sophie Bostock (Sophie@sleepio.com)
Russell Foster (russell.foster@eye.ox.ac.uk)
Vanashree Wadekar (vanashree.wadekar@psych.ox.ac.uk)
Christopher Hinds (Chris.hinds@psych.ox.ac.uk)
Colin Espie (colin.espie@ndcn.ox.ac.uk)

Version: 2
Date: 11 February 2015

Author’s response to reviews:

Thank you for the list of changes necessary.

With kind regards, Daniel